Author's response to reviews

**Title:** Occurrence of HSV-1-induced pneumonitis in patients under standard immunosuppressive therapy for rheumatic, vasculitic, and connective tissue disease.

**Authors:**

Matthias N. Witt (matthias.witt@med.uni-muenchen.de)
Gerald S. Braun (gbaun@ukaachen.de)
Stephan Ihrler (stephan.ihrler@med.uni-muenchen.de)
Holger Schmid (holger.schmid@lrz.uni-muenchen.de)

**Version:** 4 **Date:** 11 May 2009

**Author's response to reviews:**

Dear BioMed Central Editorial Production Team,

all requested formatting changes have been made according to your suggestions and all authors have proof-read the manuscript for typographical errors.

As the provided Tables (Table 1 and 2) were generated using the landscape format, we were not able to integrate them into the main manuscript body. However, we believe that both Tables would be best included into the main manuscript body of the final BMC-formatted version of the manuscript that will be published.

Thank you very much for your considerations

On behalf of the authors

Holger Schmid MD, MSc